Author's response to reviews

Title: Hispanic physicians' tobacco intervention practices: a cross-sectional survey study

Authors:
- Francisco G Soto Mas (fsotomas@hsc.unt.edu)
- Richard L Papenfuss (dpapenfuss@aol.com)
- Holly E Jacobson (hollyej@flash.net)
- Chiehwen Ed Hsu (edhsu@umd.edu)
- Ximena Urrutia-Rojas (xurrutia@hsc.unt.edu)
- William M Kane (kane@unm.edu)

Version: 5 Date: 19 October 2005

Author's response to reviews:

RESPONSE TO REVIEWERS’ COMMENTS (3)

Dear editor,
Thank you again for your interest in our manuscript and for the positive review. Please find below a description of the final changes that were made according to reviewers' comments. We have also updated authors' information and reformatted references, and figure and table legends according to the formatting checklist.

Major Compulsory Revisions
N/A

Minor Essential Revisions. We made all the changes recommended by the reviewer:
N/A

Discretionary Revisions
1. Page 5, 1st paragraph, last sentence, delete "and who can communicate with them."
   We made the change as recommended
2. Page 6, Methods, 2nd paragraph, 1st sentence, "pilot-tested."
   We made the correction as recommended